Author's response to reviews

Title: Prognostic significance of urokinase plasminogen activator and plasminogen activator inhibitor-1 mRNA expression in lymph node- and hormone receptor-positive breast cancer

Authors:

Philippe Leissner (Philippe.leissner@eu.biomerieux.com)
Thibault Verjat (Thibault.verjat@eu.biomerieux.com)
Thomas Bachelot (Bachelot@lyon.fnclcc.fr)
Malick Paye (Malick.paye@eu.biomerieux.com)
Alexander Krause (Alexander.krause@eu.biomerieux.com)
Alain Puisieux (Puisieux@lyon.fnclcc.fr)
Bruno Mougin (Bruno.mougin@eu.biomerieux.com)

Version: 5 Date: 29 August 2006

Author's response to reviews:

Dear Editorial team,
As requested, we have modified the number of the figures named now Figures 1, 2, 3, 4, 5 and 6. Modifications in the manuscript have been made accordingly. New versions of the manuscript and of the figures have been uploaded.

Best regards,
Philippe Leissner